Author's response to reviews

Title: Nerve tolerance to high-dose-rate brachytherapy in patients with soft tissue sarcoma: a retrospective study

Authors:

Tadahiko Kubo (kubot@hiroshima-u.ac.jp)
Takashi Sugita (sugita21@hph.pref.hiroshima.jp)
Shoji Shimose (shimose@hiroshima-u.ac.jp)
Toshihiro Matsuo (tmatsu@hiroshima-u.ac.jp)
Ken Hirao (hirao@ms10.megaegg.ne.jp)
Hiroaki Kimura (luna@hiroshima-u.ac.jp)
Masahiro Kenjo (kenjom@hiroshima-u.ac.jp)
Mitsuo Ochi (ochim@hiroshima-u.ac.jp)

Version: 3 Date: 11 July 2005

Author's response to reviews: see over
July 11, 2005

Dr Iratxe Puebla
Senior Editor
BMC Cancer

Re: MS: 2779817895959240

Dear Dr Iratxe Puebla,

Thank you for your e-mail of July 9, 2005. We are very pleased to hear that our manuscript was accepted for publication in BMC Cancer.

Attached is a revised manuscript by Kubo T et al. entitled “Nerve tolerance to high-dose-rate brachytherapy in patients with soft tissue sarcoma: a retrospective study” We have proofread the manuscript and have made all formatting changes as requested.

Sincerely yours,

Tadahiko Kubo, M.D.
Department of Orthopaedic Surgery
Graduate School of Biomedical Sciences, Hiroshima University
1-2-3 Kasumi, Minami-ku, Hiroshima, 734-8551, Japan.
Tel: 81-82-257-5232, Fax: 81-82-257-5234
E-mail: kubot@hiroshima-u.ac.jp